Author’s response to reviews

Title: The effectiveness and cost-effectiveness of strength and balance Exergames to reduce falls risk for people aged 55 years and older in UK assisted living facilities: A multi-centre, cluster randomised controlled trial

Authors:

Emma Stanmore (emma.stanmore@manchester.ac.uk)
Alexandra Mavroeidi (Alexandra.mavroeidi@strath.ac.uk)
Lex de Jong (lex.dejong@curtin.edu.au)
Dawn Skelton (dawn.skelton@gcu.ac.uk)
Chris Sutton (chris.j.sutton@manchester.ac.uk)
Valerio Benedetto (VBenedetto@uclan.ac.uk)
Luke Munford (luke.munford@manchester.ac.uk)
Wytske Meekes (W.M.A.Meekes@uvt.nl)
Vicky Bell (vicky.bell@manchester.ac.uk)
Chris Todd (chris.todd@manchester.ac.uk)

Version: 1 Date: 12 Oct 2018

Author’s response to reviews:

Thank you for the opportunity to correct the uploading error in this manuscript, please find the correct Table 3 now added and the Figure 1 removed from the manuscript,

Best wishes,

Dr Emma Stanmore